Author’s response to reviews

Title: Rate and determinants of treatment response to different antiretroviral combination strategies in subjects presenting at HIV-1 diagnosis with advanced disease.

Authors:

Antonella Esposito (antonellaesposito2002@yahoo.it)
Marco Floridia (marco.florida@iss.it)
Gabriella d’Ettorre (gabriella.dettorre@uniroma1.it)
Daniele Pastori (daniele.pastori@yahoo.it)
Alessandra Fantauzzi (alessandrafantauzzi@yahoo.it)
Paola Massetti (paola.massetti@uniroma1.it)
Giancarlo Ceccarelli (giancarlo.ceccarelli@uniroma1.it)
Camilla Ajassa (camilla.ajassa@uniroma1.it)
Vincenzo Vullo (vincenzo.vullo@uniroma1.it)
Ivano Mezzaroma (ivano.mezzaroma@uniroma1.it)

Version: 4 Date: 12 December 2011

Author’s response to reviews:

In Abstract (Results section) line 5 delete the words "protease inhibitors and the square brackets

Table1: Baseline CD4(patients’ percentages have been rounded to the nearest integer); HIV RNA (patients’ percentages have been rounded to the nearest integer) Footnote (HBV, HCV and HBV+HCV percentages have been rounded)

In Table 3 footnote please change as follow: * for unboosted PIs and ** for boosted PIs

Q1 OK
Q2 OK